

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Gould, Eric D.; Lavy, Victor; Paserman, Marco Daniele

Working Paper Does immigration affect the long-term educational outcomes of natives? Quasi-experimental evidence

IZA Discussion Papers, No. 1883

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Gould, Eric D.; Lavy, Victor; Paserman, Marco Daniele (2005) : Does immigration affect the long-term educational outcomes of natives? Quasi-experimental evidence, IZA Discussion Papers, No. 1883, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/33288

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 1883

Does Immigration Affect the Long-Term Educational Outcomes of Natives? Quasi-Experimental Evidence

Eric D. Gould Victor Lavy M. Daniele Paserman

December 2005

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Does Immigration Affect the Long-Term Educational Outcomes of Natives? Quasi-Experimental Evidence

Eric D. Gould

Hebrew University CEPR and IZA Bonn

Victor Lavy

Hebrew University CEPR and NBER

M. Daniele Paserman

Hebrew University CEPR and IZA Bonn

Discussion Paper No. 1883 December 2005

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 Email: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit company supported by Deutsche Post World Net. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

IZA Discussion Paper No. 1883 December 2005

ABSTRACT

Does Immigration Affect the Long-Term Educational Outcomes of Natives? Quasi-Experimental Evidence^{*}

This paper uses the mass migration wave to Israel in the 1990s to examine the impact of immigrant concentration during elementary school on the long-term academic outcomes of native students in high school. To identify the causal effect of immigrant children on their native peers, the empirical strategy must address two sources of bias: the endogenous sorting of immigrants across schools, and the endogenous grade placement of immigrants within schools. We control for the endogeneity of immigrant placement across schools by conditioning on the total number of immigrants in a school and exploit random variation in the number of immigrants across grades within the same school. To address the endogenous grade placement of immigrants within schools, we use the immigrants' dates of birth as an instrument for their actual grade placement. The results suggest that the overall presence of immigrants in a grade had a significant and large adverse effect on two important outcomes for Israeli natives: the dropout rate and the chances of passing the high school matriculation exam which is necessary to attend college.

JEL Classification: I20, J24

Keywords: school quality, natural experiment, peer effects, dropout rates, immigrant absorption

Corresponding author:

M. Daniele Paserman Department of Economics Hebrew University Jerusalem, 91905 Israel Email: dpaserma@shum.huji.ac.il

This is a substantially revised version of Gould, Lavy, and Paserman (2004b). We thank Joshua Angrist, Esther Duflo, Caroline Hoxby, Chris Taber and seminar participants at the Hebrew University, Bocconi University, the Bank of Israel, Ben-Gurion University, LSE, the NBER Education Meetings, the University of Helsinki, the Institute for International Economic Studies in Stockholm, Northwestern University, the University of Chicago Graduate School of Business, UC Berkeley, Stanford University, the conference on "Inequality, Minorities and Social Exclusion" at Bar-Ilan University, the CEPR conference on "Inequality in Labour Markets," and the European Summer Symposium in Labor Economics. Vadim Krimsky and Tamar Roth provided expert research assistance. All errors are our own.

1. Introduction

An increasingly important issue faced by Western countries at the turn of the 21st century is the social tension created by growing numbers of immigrants from less developed countries. The effect of immigration on the local labor market has received considerable attention in the literature, but little is known about the impact of immigration on the school system. This examines the impact of immigrant concentration during elementary school on the long-term academic outcomes of native students in high school. The focus on the long-term outcomes of natives, seven years after their exposure to the "treatment," is one of the distinguishing features of this paper.

Our analysis exploits a unique opportunity to identify peer effects of immigrant children in the classroom using the mass migration wave from the former Soviet Union ("FSU") to Israel in the early 1990's. The sheer size of the immigration wave, nearly 20 percent of the total population within a decade, produced large variation in the absorption level of recent immigrants across schools throughout the country. The raw data show that native students who went to school with more immigrants in 5th grade typically had lower achievements in high school. This correlation is not surprising, given that immigrants tended to locate in poorer areas where housing costs were lower. Therefore, identification of the causal effect of immigrants on their native peers must account for the endogenous placement of immigrants into areas that are more likely to be populated with lower-achieving native students, regardless of the local level of immigrant concentration.

In addition, the data reveal that immigrant students are much more likely to be held back a grade than native students. For example, only 55 percent of the immigrant children in our sample who are supposed to be in 5^{th} grade according to their date of birth are actually in the 5^{th} grade. The decision to hold an immigrant student back a grade within a school could be influenced by the characteristics of the native students in the two adjacent grades. It could be the case that immigrant students were more likely to be placed with the stronger or weaker students within the same school, thus producing a spurious correlation between the presence of immigrants in the grade and the future outcomes of native students. Therefore, identification of the causal effect of immigrants on their native peers must correct for two potential sources of bias: the endogenous sorting of immigrants across schools (mostly into poorer areas), and the endogenous grade placement (holding back) of immigrants within schools.

To control for the endogeneity of immigrants across schools, our empirical strategy exploits random variation in the number of immigrants across grades within the same school. For example, conditional on the total number of immigrants in a given school in grades 4 to 6, the actual number of immigrants in grade 5 can be considered as being determined solely by random factors such as variation in the year of birth among the pool of immigrant children in the school district. This natural experiment is best thought of as the random drawing of balls from an urn in which one third of the balls are marked with the number "4", one third with the number "5", and one third with the number "6". Conditional on the total number of balls drawn, the number of "5" balls is as good as randomly assigned. Conditioning on the total number of immigrants in the advanced grades in elementary school removes much of the heterogeneity across schools that may confound our estimates: schools that absorbed the same number of immigrants, but also in other characteristics.

To account for the second type of selection, the potentially endogenous holding back of immigrants within a school, as well as to address any potential attenuation bias due to measurement error in the grade placement of students in our administrative data, we instrument for the actual percentage of immigrants in 5th grade with the *predicted* percentage of immigrants. The predicted percentage is calculated by assigning to each child the grade he/she should have attended based on his/her exogenously determined date of birth. Our identifying assumption in this case is that the predicted number of immigrants in grade 5 is as

good as random, conditional on the *predicted* number of immigrants in grades 4 to 6. Summing up, the empirical strategy controls for the endogenous placement of immigrants across schools by conditioning on the total number of immigrants in grades 4 to 6, and controls for the endogenous grade placement of immigrants within schools by using their dates of birth as an instrument for their actual grade placement.

To implement our empirical strategy, we use administrative panel data on school enrollment and test scores for each 5^{th} grade child from the 1993-1994 school year until the 2000-2001 school year, the year this cohort was scheduled to graduate from high school. A set of balancing tests indicate that both the actual percentage of immigrants and the predicted percentage of immigrants in 5^{th} grade are not systematically related to school and student background characteristics, once we control for the number of immigrants in grades 4 to 6 and the total number of children in grade 5. When we regress the high school achievements of native students on the *actual* percentage of immigrants in 5^{th} grade, we find that the concentration of immigrants in elementary school has essentially no effect on their dropout rates, while it has a relatively small and marginally significant effect on high school matriculation rates. In contrast, using the instrumental variable strategy to account for the endogenous placement of immigrants across grades, we find larger and more significant adverse effects of the presence of immigrants on the outcomes of natives.

We also show that high school achievements of children who were in 5th grade in 1994 are affected only by the percentage of immigrants in 5th grade, but not by the percentage of immigrants in first and second grades. This lends strength to the notion that our estimates are indeed picking up the effect of immigrant concentration in the grade, and not the effect of other unobserved school characteristics. Finally, we look separately at the effects of immigrant boys and immigrant girls on the achievements of native boys and girls. We find that native children are especially affected by the presence of immigrant children of their own

3

gender, suggesting that part of the mechanism driving the effect may be due to the influence of peers.

The rest of the paper is structured as follows: the next section reviews the existing literature that is most closely related to our work. Section 3 presents a brief background of the immigration episode we are exploiting and describes our data. Section 4 describes the empirical methodology for dealing with the endogenous placement of immigrants across schools. Section 5 adapts the strategy in Section 4 in order to generate instrumental variables (IV) results that take care of the endogeneity of immigrants across schools and the potential endogenous placement of immigrants across grades within a school. Section 6 performs robustness checks and Section 7 examines the gender composition of the immigrants and discusses the magnitude of the effects. Section 8 concludes.

2. Literature Review

While there is a vast literature on the effects of immigration on native labor market outcomes,¹ the question of whether immigration affects natives' educational outcomes has received relatively little attention. Exceptions include Betts (1998), who examines whether immigration reduces the contemporaneous high school graduation rate of natives, and Hoxby (1998) and Borjas (2004), who look at whether immigrants crowd-out natives from slots in college and graduate programs. Also, Betts and Fairlie (2003) investigate whether immigration in California induced native flight from public to private schools.

Closely related to our paper is also the literature on desegregation and classroom peer effects: Angrist and Lang (2004) study the effects of Metco, a long-running desegregation program that sends minority children out of the Boston public school district to a large and affluent suburban district. Overall, they find little evidence that the test scores of non-Metco

¹ See for example Card (1990, 2001); Altonji and Card (1991), Lalonde and Topel (1991), Hunt (1992), Borjas, Freeman and Katz (1996), Borjas (2003); in the Israeli context, see Friedberg (2001) and Cohen-Goldner and Paserman (2004 and forthcoming).

students are affected by the presence of disadvantaged peers, although there seems to be some evidence of a negative effect on the test scores of non-Metco minority students. Guryan (2004) studies the effects of desegregation from the 1950s to the 1980s, and concludes that black dropout rates fell as a result of desegregation, while desegregation did not affect the dropout rates of whites. Other studies that have looked directly at peer effects in the classroom using large administrative data sets include Boozer and Cacciola (2001), Hanushek et al. (2003), Lefgren (2004), Betts and Zau (2004), Vigdor and Nechyba (2004) and Burke and Sass (2004). Sacerdote (2001) and Zimmermann (2003) study peer effects in a university setting, and find that academic and social outcomes of college students are affected by the quality of randomly assigned roommates. Hoxby (2000a) uses an identification strategy similar to our own, exploiting idiosyncratic variation in gender and race composition of adjacent cohorts in Texas public schools, and in a wide range of specifications she finds that children's elementary school test scores are affected by those of their peers, with intra-race peer effects appearing to be particularly strong.² In contrast to Hoxby, our focus is on the effect of immigrants on natives and we examine long-run effects, rather than the contemporaneous effects of peers.

Recent years have also seen a flurry of research that uses exogenous variation created by true or natural experiments to study the effects of peers and the environment at the neighborhood level on a variety of educational, health, and economic outcomes (Katz, Kling and Liebman, 2001; Edin, Fredriksson and Åslund, 2003; Oreopoulos, 2003; Jacob, 2004; Weinberg, Reagan and Yankow, 2004). In the Israeli context, Gould, Lavy and Paserman, (2004a) found that Ethiopians who arrived during Operation Solomon generally benefited from attending better schools. However, before claiming that sending immigrants to better-performing schools represents a net gain from the point of view of society, it is crucial to

 $^{^{2}}$ An identification strategy based on random variation in cohort size at the school level is also used by Angrist and Lavy (1999) and Hoxby (2000b) to estimate the effects of class size on student achievements.

assess whether children in immigrant or minority-receiving schools are affected by the influx of disadvantaged peers, as we do in the current study.

3. Background and Data

The mass migration from the former Soviet Union to Israel since 1989 can be viewed as a unique laboratory for the social sciences. More than one million immigrants moved to Israel since 1989, increasing its population by a remarkable 20 percent within a decade. The migration wave was also highly concentrated, with approximately 330,000 immigrants arriving in 1990 and 1991. The fall of the Berlin wall in November 1989 was the trigger that started the whole process, as a number of factors combined to induce migration. First, the Soviet Union lifted its emigration restrictions. Second, the political instability and the uncertain economic climate in the beginning of the 1990s greatly increased the incentives to migrate. Third, Israel was one of the few viable options, as it imposed no entry restrictions and no lengthy waiting periods. Along with the mass migration from the former Soviet Union, the early 1990s also saw the exodus of 15,000 Ethiopian Jews in the context of Operation Solomon in May 1991, and increasing immigration rates from other countries, mainly the United States and Argentina.

Not surprisingly, there are large differences in the family background of immigrants from different source countries, and the overall immigrant population is quite different from native Israelis in terms of family background. These differences are displayed in panel A of Table 1, which shows the background characteristics of 5th grade students in 1994, by immigrant status and by country of origin, based on the Ministry of Education administrative data that is used throughout the paper. We define as immigrants all children who were born outside of Israel and immigrated after January 1st, 1989. The table reveals that immigrant children from the former Soviet Union generally have parents who are more educated than native Israeli children (a little more than one additional year of education for each parent), and

come from smaller families (1.10 siblings versus 2.41 for native Israelis). In contrast, Ethiopian immigrant children come from much larger families (an average of 4.28 siblings) but have parents with only one or two years of education. The vast majority of immigrants during this period came from the former Soviet Union, so the characteristics of the overall immigrant population are dominated by the characteristics of the Soviet immigrants.

While it appears that native Israeli children were exposed to a new immigrant population with more favorable family background characteristics, it is also true that the new immigrant population displayed several signs of social and economic distress. Panel B of Table 1 presents characteristics of households with children between the ages of 8 and 12 drawn from the 1995 Israeli Census.³ Compared to native Israeli children, Table 1 reveals that immigrant children came from poorer households which tend to rent rather than own their own homes, and where the head of the household is much more likely to be female and/or unemployed. Again, the Ethiopian immigrants display much larger signs of weakness compared to the Russians, but both groups of immigrants are less favorable compared to the native Israeli population. So, while it is true that the overall immigration wave during this period flooded the country with immigrants who were generally well-educated, the overall picture reveals that these immigrants displayed several signs of socio-economic difficulties, which made them weaker in many respects than the native population. Therefore, despite the unique characteristics of this mass migration episode, immigrants to Israel share characteristics that are common to immigrants in all Western societies, making our results relevant beyond the Israeli context.

To examine whether the influx of this immigration wave into elementary schools influenced the high school outcomes of native Israeli children several years later, we link detailed information on each child's school environment in the 5th grade to his or her achievements throughout high school. The data comes from administrative records collected

³ The "long questionnaire" of the Israeli Census was administered to 20% of Israeli households.

by Israel's Ministry of Education on the universe of Israeli elementary schools in the 1993-1994 school year, the first year a computerized file is available. The data is based on reports from school authorities to the Ministry of Education at the beginning of the school year. The file contains an individual identifier, a school and class identifier, and detailed demographic information on all immigrant pupils in grades 1 to 6, and all pupils in grades 1, 2, and 5.⁴ We are able to exactly identify pupils by their country of origin and date of immigration (month and year), so we know precisely who is a native Israeli student and how many immigrant children were present in the 5th grade class of each elementary school. We focus our attention on the 5th grade cohort in 1993-1994 for two reasons. First, this is one cohort for which we have complete data on native Israeli pupils. Second, 5th grade students in the 1993-1994 school year had enough time within our sample period (which ends with the 2000-2001 school year) to finish high school if they progressed through the system without repetition. The data set also contains the birth date of each child, which is mapped into each child's *predicted* grade level using the enrollment cut-off dates for that cohort.⁵ The predicted grade level will be used in our instrumental variable strategy.

We link the elementary school records to individual data on high school enrollment and matriculation exam outcomes in the 1998-1999, 1999-2000, and 2000-2001 school years. Therefore, we are able to follow each native Israeli student from 5th grade in 1993 all the way through the advanced stages of high school. We study two important high school outcomes: dropping out before completing 12th grade, and passing the matriculation exams. The latter outcome is particularly important in Israel since it is required to attend college. Similar high school matriculation exams are found in many countries and in some states in the United

⁴ Note that we do not have any data on native children enrolled in grades 4 and 6 in 1993/1994. This prevents us from using a traditional identification strategy based on multiple cohorts and school fixed effects. In practice, though, the estimator we use is very similar to a fixed effects estimator.

⁵ In Israel, children roughly enter first grade in September of the calendar year in which they turn six years old. We say "roughly" because the relevant threshold dates are based on the Hebrew calendar. For example, the first grade class of September 2005 is composed of children born between the 1st of Tevet 5759 (December 20th, 1998) and the 30th of Kislev 5760 (December 9th, 1999). For conversion between Hebrew dates and Gregorian dates, see http://www.hebcal.com.

States. Examples include the French Baccalaureate, the German Certificate of Maturity (Reifezeugnis), the Italian Diploma di Maturità, the New York State Regents examinations, and the recently instituted Massachusetts Comprehensive Assessment System.

We should note that our key variable of interest, exposure to immigrants in 5^{th} grade, is taken as a proxy for the total exposure to immigrants throughout elementary school. By the end of 5^{th} grade, native students had been exposed to immigrants for an average of 31.7 school months (slightly more than 3 school years). Interestingly, the length of exposure to immigrants is nearly completely uncorrelated with the proportion of immigrants in the school.

Table 2 presents summary statistics for various measures of the 5th grade learning environment faced by native Israeli students. We classify the learning environment variables into two groups, "peer characteristics" and "school characteristics." The "peer characteristics" describe the family background of the native Israeli 5th grade students in the school. The "school characteristics" contain additional aspects measured at the school level: average scores for the 4th and 5th grade students on the 1991 standardized math and verbal tests⁶ and a 1991 socio-economic index of students in each elementary school.⁷ All of the "school characteristics" are taken from the 1991 school year, which is two years before the sample of native students used in Table 2 attended 5th grade.

Summary statistics are presented in four columns in Table 2, whereby the sample used in each column is defined by the total number of immigrants absorbed into the elementary school in grades 4 to 6. For example, the first column presents summary statistics for only the native Israeli 5th graders who attended an elementary school in 1993-1994 where the total

⁶ In June of 1991, near the end of the school year, all fourth and fifth grade students were given achievement tests designed to measure mathematics and Hebrew reading skills. The scores used here consist of the average score on some of the basic and all of the more advanced questions in the test, so that scores are scaled from 1 to 100. The low average and the high variation of scores on this achievement test generated considerable public controversy in Israel, leading to the abandonment of the national testing program in 1992. This is the same data used in Angrist and Lavy [1999].

⁷ The socioeconomic index is based on a function of the pupils' fathers' education and continent of birth, and of family size. The raw index is recorded as the percentage of students in the school who come from what is defined to be a disadvantaged background. Our socioeconomic status index is obtained by standardizing this variable and multiplying it by -1, so that high socioeconomic status schools have a high value of the index.

number of post-1989 immigrants in grades 4 to 6 was between 1 and 10. The next column uses a range of immigrants in 4th to 6th grades between 11 and 20, and the next two columns use ranges of 21-30 and 31-40 respectively. The last column shows the summary statistics for all schools that had between 1 and 40 immigrants in grades 4 to 6.

A close inspection of the numbers in Table 2 reveals why it is important to break down the sample in this manner. When comparing the means across columns, a clear pattern emerges: schools with larger numbers of total immigrants in grades 4 to 6 are increasingly worse off in terms of the characteristics of the native Israeli students. For example, as the number of immigrants in the school increases, the parental education of native students falls, the percent of students with origins from Asia and Africa (which is widely considered to be a disadvantaged ethnic group in Israel) increases, the average math and verbal scores of the elementary school fall, and the socio-economic index of the school falls. In addition, immigrants represent only 4.04 percent of the 5th grade class when the total number of immigrants in 4th to 6th grades is between 1 and 10, but increases to 15.47 percent when there are 31-40 immigrants in the same three grades. Overall, a clear pattern emerges: larger numbers of immigrants in grades 4 to 6 are associated with larger concentrations of immigrants in 5th grade, weaker native Israeli students in terms of their family background, and weaker schools in terms of student test scores in previous years. These results are not very surprising since new immigrants were not placed or directed to areas where the best schools and most-educated parents are likely to be. Instead, immigrants typically started out in areas with lower-cost housing, and therefore native Israelis from disadvantaged backgrounds and lower-performing schools were more likely to attend schools with a higher concentration of immigrants.

The lower panel of Table 2 presents the high school outcomes of native Israeli students according to the number of immigrants in grades 4-6 in their school in 1993-1994, while they were in 5th grade. Not surprisingly, the table indicates that native students who attended

schools with a larger proportion of immigrants in 5th grade also had lower outcomes in high school: they dropped out of high school more often, and were less likely to pass the matriculation exams. The first column in Table 3 demonstrates these correlations more clearly by regressing each outcome variable on the percent of immigrant students in each native child's 5th grade class. For both outcomes, the coefficient on the proportion of immigrants in 5th grade is large and highly significant. A 10 percentage point increase in the immigrant share is associated with a 1.3 percentage point increase in the dropout rate, and a 5.7 percentage point decrease in the matriculation rate. However, the coefficients are reduced dramatically in size and significance in the second column of Table 3, which controls for the individual and school characteristics described in Table 1. Adding an additional control for the number of immigrants in the school from grades 4 to 6 in column (3) in Table 3 has little effect on the size of the coefficients, but it increases the standard errors and, as a result, the coefficient on the dropout rate becomes insignificant.

Overall, the results in Tables 2 and 3 show that natives from weaker backgrounds tend to be with more immigrants in 5th grade, and that the proportion of immigrants in 5th grade is negatively correlated with natives' high school outcomes. As a result, we cannot draw any inferences about causality since natives who attended elementary schools with more immigrants may have achieved less in high school because of the presence of the immigrants, or because of other characteristics that are correlated with a larger concentration of immigrants. Although we can control for many observable characteristics which explain much of the strong correlation between the proportion of immigrants and high school performance in Table 3, it is likely that immigrant concentration is correlated with unobservable characteristics of the native students and their schooling environment, given that it is so strongly correlated with observable characteristics of the students and the schools they attended. For this reason, we call the estimates in Table 3 "naïve" in the sense that they are

likely to suffer from an omitted variable bias. In the next section, we develop a strategy to overcome this identification problem.

4. OLS Stratification: Identification Strategy and Results

To estimate the causal effect of the presence of new immigrants on their native Israeli peers, we need variation in the presence of immigrant children in the classroom that is uncorrelated with potential outcomes of the native children. It is clear from Table 2 that immigrant concentration is not randomly assigned – native students in poorer and lower achieving schools are more likely to attend schools with larger concentrations of immigrant students. In general, the non-random assignment of children to schools, either because of the choice of schools exercised by the families, or because of the endogenous matching of pupils to schools done by education authorities, makes it difficult to identify the effect of one group of peers on the outcomes of others. As such, the identification of a causal link between peers has proven to be a difficult task in the literature.

To overcome this obstacle, our strategy is to exploit random variation in the number of immigrants in 5^{th} grade conditional on the school absorbing a given range of immigrant students in grades 4 to 6. Initially, we assume that the *actual* number of immigrants in grade 5 is random, conditional on the actual number of immigrants in grades 4 to 6. This allows us to state clearly the assumptions needed for identification and to describe in detail the estimation strategy in the simple case where the placement of immigrants *across* grades *within* a given school can be thought of as random. To account for selective grade placement of immigrants within a school, as well as to address potential attenuation bias due to measurement error in our administrative data, we will later employ a second identification strategy, which uses the number of immigrants that should have attended 5th grade based on their birth dates as an instrument for the actual number of immigrants.

Conditional Random Assignment of the Actual Number of Immigrants

Formally, we define the treatment variable t as the percentage of immigrant students in grade 5 in a given school:

$$t = 100 \times \frac{\text{Number of immigrants in grade 5}}{\text{Number of children in grade 5}}$$
.

We have a random sample of units, indexed by i=1,2,...,N. For each unit, we postulate the existence of a potential outcome, $Y_i(t)$, for $t \in T \equiv [0,100)$. We observe for each unit the actual treatment T_i and a set of covariates X_i . The observed outcome for unit *i* is $y_i^{obs} = \int_{t \in T} \mathbf{1}(T_i = t)Y_i(t)dt$.

Assumption 1 (Conditional random assignment):

Y(t) independent of $T \mid X$, for all $t \in [0, 100)$.

Assumption 2 (Constant linear treatment effect):

$$Y_i(t) = Y_i(0) + \beta t$$
.

The first assumption is at the heart of our strategy to account for the non-random placement of immigrants across schools. In our case, the critical conditioning covariates are the number of immigrant children in grades 4 to 6, and the total number of children in grade 5. We know that schools that absorbed large numbers of immigrants are different in terms of student and school characteristics relative to schools that absorbed fewer immigrants. However, holding constant the number of immigrant children in grades 4 to 6, the number of immigrants in grade 5 (the numerator in the treatment variable) is determined solely by random variation in the grade distribution among the pool of immigrant children in a school district. Since the treatment variable is the immigrant *ratio*, we must also condition on the total number of children in grade 5. Our strategy is based on the assumption that among schools of the same size that absorbed the same number of immigrant children in grades 4 to 5.

6, variation in the percentage of immigrant students in grade 5 is determined primarily by random variation in the size of each immigrant cohort.

We now consider the expectation of the observed outcome, conditional on the actual treatment received and on the covariates:

$$E(y^{obs} | T = t, X) = E[Y_i(0) | T = t, X] + \beta t$$
$$= E[Y_i(0) | X] + \beta t$$

where the first equality follows from the linearity assumption, and the second equality follows from the conditional random assignment assumption. The above formulation illustrates that, conditional on any particular value of the covariates, we can obtain a consistent estimate of the treatment effect β by running a regression of the observed outcome on a constant and the treatment variable.

In practice, this approach is not feasible, because the two key conditioning covariates are essentially continuous, so that there are not enough observations for any given value of *X*. For example, there are few native students who attended schools with, say, 70 total students in grade 5 and 15 total immigrants in grades 4 to 6. Therefore, we divide the sample into eight different intervals defined by the number of immigrants in grades 4 to 6 in a given school: the first interval includes schools with 1 to 5 immigrants in grades 4 to 6, the second interval includes schools with 6 to 10 immigrants in grades 4 to 6, and so on. Then, within each interval, we regress the outcome variable on the treatment and on a flexible function of the key conditioning variables. This *control* function, as it is known in the literature,⁸ is meant to capture the term E[Y(0)|X = x] in equation (1). In practice, the control function will include the number of immigrants in grades 4 to 6, the squares of these two variables, the interaction between the two, and, in some specifications, additional individual and school characteristics entered linearly. Note that since the exact functional form of E[Y(0)|X = x] is unknown, estimating equation (1) on the full sample imposes the

specific functional form assumption globally, and may be misleading. Instead, by dividing the sample into narrow intervals on the basis of the number of immigrants in grades 4 to 6, we are less likely to induce bias in our estimates by our local approximation to E[Y(0)|X = x].

Summing up, within each interval we run the following regression to explain the high school outcome of native Israeli student *i* who attended elementary school *j*:

(High School Outcome)_{ij} =
$$\lambda_0 + \beta$$
 (Percent Immigrants in 5th Grade)_j
+ λ_1 (Family Background)_i
+ f (# of immigrants in grades 4 to 6, # of children in grade 5)_j
+ u_{ij} ,

where the treatment parameter, β , is identified by the conditional random assignment assumption described above. Since the percent of immigrants in 5th grade does not vary at the individual level, the standard errors are adjusted for clustering at the elementary school level. The average treatment effect can then be calculated by simply taking the weighted average of the coefficients within each interval. Alternatively, one can pool all intervals together, and then regress the outcome variable on the treatment effect, a dummy variable for each one of the intervals, and the interactions of this dummy with each one of the conditioning variables. This specification is equivalent to running separate regressions within each sub-sample, but constraining the coefficient on the treatment variable to be the same across sub-samples. This allows us to pool together the data from the whole sample and obtain one single estimate of the treatment effect. In the tables that follow, we refer to this model as the "fully-interacted" model.

Balancing Tests

To determine whether the data supports our identification strategy, we now test whether there is a significant relationship between immigrant concentration in grade 5 and the

⁸ Heckman and Robb (1985).

observable characteristics of native Israeli children in the various sub-samples of schools defined by the total number of immigrant children in grades 4 to 6. This is done by regressing various measures of the native student's background or elementary school environment on the percent of immigrant students in the school's 5th grade for the eight sub-samples. If the assignment of immigrant students across the 4th, 5th and 6th grades within the same school is indeed random once we condition on the narrow range of total immigrants in the school and on the control function, then we should find no such significant relationship.

This type of balancing test does not necessarily provide a proof of random assignment, as the assumption requires there to be no correlation between the percentage immigrants in 5^{th} grade and both observable and *unobservable* background characteristics. However, the lack of a significant relationship between the percentage of immigrants in 5^{th} grade and observable characteristics are also unlikely to be correlated with the percentage of immigrants in 5^{th} grade. The results of these tests are presented in the first eight columns of Table 4.

Each coefficient in Table 4 represents the estimate from a single regression for each of the various measures of the native student's background and elementary school environment on the percent of immigrants in 5th grade.⁹ Out of the 64 coefficients, only six are significant at the ten percent significance level or lower. This is almost exactly the rejection rate that we would observe under the null hypothesis that the covariates are indeed balanced. The other 58 coefficients point to no systematic relationship between observable characteristics and the percent of immigrants in 5th grade. Furthermore, the six coefficients which do turn out to be statistically significant do not tell a consistent story about whether students are positively or negatively selected into schools with larger percentages of immigrants in the 5th grade (holding constant the range of immigrant students in grades 4 to 6). For example, the second column reveals that higher proportions of immigrants in 5th grade are significantly associated

with larger families and lower class sizes, which leads to contradicting inferences about negative or positive selection. By contrast, a higher proportion of immigrants is significantly related to smaller families and larger class sizes in the third column. These coefficients not only have contradictory implications for the direction of selection within a given sub-sample, but they also yield contradictory implications for the direction of selection for a given variable across sub-samples. The bottom row of the table shows that, within cells, the standard deviation of percentage immigrants in 5th grade is fairly large, implying that our failure to find significant coefficients is not due to insufficient variation in the explanatory variable.

The next column in the table reports the average of the eight coefficients for each row, weighted by the inverse of the variance of the coefficients,¹⁰ and its associated t-statistic. This column confirms that there is no consistent or significant relationship between the observable characteristics of native students and the percentage of immigrants in 5th grade, once the number of immigrants in grades 4 to 6 is limited to a narrow range. The last column in the table reports the "naïve" OLS coefficient from a simple regression of each variable on the percentage of immigrants in grade 5 in the entire sample. Contrary to the coefficient estimates within each sub-sample, these coefficients are large and statistically significant, and point to a strong negative relationship between immigrant concentration and the background characteristics of natives and schools. The contrast between the "naïve" OLS coefficients and the coefficients within the narrow cells illustrates the extent to which our identification strategy can reduce bias stemming from the non-random selection of immigrants across schools.

⁹ The regressions are at the individual student level, with standard errors adjusted for clustering at the school level.

¹⁰ This is essentially a minimum distance estimator, where each one of the eight coefficients is a separate estimate of the common parameter of interest.

The OLS-Stratified Regression Results

Table 5 presents the estimates of the long-run effects of immigrants on the high school achievements of native Israeli students, using regressions within each stratified sub-sample as described above. Each entry in the table represents the coefficient estimate from a separate regression on the treatment variable (the percent of students in 5th grade who are recent immigrants) for the two high school outcomes of interest: the probability of dropping out before completing 12th grade, and the probability of receiving a matriculation diploma at the end of high school. The estimates reported in the upper panel of the table are from regressions that include only the treatment variable and the flexible control function (the number of immigrants in grades 4 to 6 and its square, the number of children in grade 5 and its square, and the interaction between these two variables). The estimates in the bottom panel come from regressions that include additional school-level variables (test scores and socio-economic index of the students in 1991, a dummy for being a religious school, and class size in 1994), as well as personal characteristics of each student (father's and mother's schooling, number of siblings, and ethnic origin). The first eight columns in the table correspond to the eight samples used in Table 4 for the balancing tests.¹¹

Most of the estimates in the top panel of Table 5 are small with large estimated standard errors, and not statistically different from zero. When the additional controls are included in the lower panel of the table, the results are similar. The fact that the coefficients do not change very much after adding personal and school-level characteristics into the specification may be interpreted as additional evidence that the treatment and control samples are relatively balanced in terms of these observable measures within each sub-sample.

Although most of the coefficients in each sub-sample are insignificant, the general pattern of the signs of the coefficients points in the direction of an adverse effect of immigrants on the high school outcomes of natives. In the last two columns of Table 5, we

present two summary estimates of the treatment effect and their associated t-statistics. The ninth column is simply the weighted average of the eight within-interval treatment effects, and the last column presents the treatment effect obtained from the fully interacted model discussed above. These last two columns indicate that immigrant concentration in elementary school has a marginally significant negative effect on high school matriculation rates, and an insignificant effect on the dropout rate. The point estimates imply that an increase of 10 percentage points in the immigrant concentration in grade 5 lowers the individual matriculation rate by 1.2-1.6 percentage points.

5. The Endogenous Distribution of Immigrants across Schools and Grades

The analysis in the previous section is based on the assumption that the endogenous placement of immigrants in the 5th grade across schools is eliminated once we control for the total number of immigrants in the school from grade 4 to 6. That is, given that a school has a particular number of total immigrants in those three grades, the number that winds up in each grade is as good as random. This assumption is supported by the balancing tests in Table 4, which showed no correlation between the observable characteristics of natives and 5th grade immigrant concentration within each stratified sample. It appears that the stratification corrects for the endogeneity problem across schools, whereby more immigrants are found in poorer areas with more disadvantaged natives. However, given that a school has a particular number of immigrants in grades 4 to 6, there is reason to be concerned that the distribution of immigrants across grades within the same school may not be entirely random. This concern is based on the high rate of holding students back a grade among immigrants.

For example, among the immigrant children who should have attended 5th grade based on their date of birth and the enrollment rules in place for that cohort, 55 percent were in fact

¹¹ The sample size in each of the columns includes only native Israeli pupils, and therefore, is identical to the sample size reported in the respective column of Table 4.

enrolled in grade 5, while 38 percent are in grade 4.¹² The remainder are scattered between grade 6 (children who have been pushed up a grade, about 3 percent), and grades 1, 2 and 3 (4 percent).¹³ This would not be a cause for concern if we could safely assume that a school's propensity to hold back immigrants is uncorrelated with differences in characteristics of native students across grades within the same school. If, however, the school tends to hold immigrant students back so that more of them are placed in the grade with better or worse native students compared to the adjacent grade, then the results in the previous section could be biased. The balancing tests indicate that there is no such endogenous placement based on observable characteristics, but the high rate of holding immigrant students back could be correlated with unobservable characteristics, such as behavioral problems of students in one grade versus the other. If true, then the previous results would be biased towards zero if immigrant students are endogenously held back in order to be with better native students, while an upward bias (in absolute value) would result from being placed with weaker native students. Measurement error in the grade enrollment information of immigrants would also bias the results in the previous section towards zero.¹⁴

An Instrumental Variable Strategy

To account for the potential endogeneity of immigrant placement within schools, we adapt our identification strategy by predicting how many immigrants are supposed to be in grade 5 within a school based on their date of birth and using this prediction to instrument for the actual immigrant concentration in grade 5 within a school. This strategy assumes that the

¹² Source: authors' calculations from the Ministry of Education data.

¹³ We cannot report analogous statistics for native children, as we have no information on natives in grades 4 and 6 in 1993/94. However we can compare the distribution of natives and immigrants who should have attended second grade: 86 percent of natives were indeed enrolled in second grade, compared to 72 percent of immigrants. ¹⁴ Even with the best administrative data, some students' grades may be recorded incorrectly, or there may be movements across grades in the middle of the school year. If measurement error is of the classical type, this would lead to attenuation bias. Moreover, attenuation bias could be exacerbated by controlling for variables such as the number of immigrants in grades 4 to 6, since the standard deviation in this latter variable is probably larger than the standard deviation in the number of grade 5 immigrants; hence, much of the true signal in the explanatory variable would be swept away by the control variable, leaving mostly noise.

predicted percentage of immigrants in grade 5 (i.e., the percentage of immigrant children who should have been in grade 5 based on their dates of birth) is as good as random, conditional on the *predicted* number of immigrants in grades 4 to 6, and on the total *predicted* number of children in grade 5. In other words, conditional on the total number of immigrant children born between December 1981 and November 1984 in a given school, the number of immigrant children born between December 1982 and November 1983 is as good as random. This strategy abstracts from endogenous holding back decisions by parents and schools, and assumes only that the distribution of immigrant children's birth dates is random, conditional on the total number of immigrants in a school.

Formally, we define Z as the predicted percentage of immigrant children in grade 5^{15} and we replace Assumption 1 with the following:

Assumption 1' (Conditional random assignment of the instrument):

Y(t) independent of
$$Z | \tilde{X}$$
, for all $t \in [0, 100)$.

Note that the conditioning variable is \widetilde{X} , the *predicted* number of immigrants in grades 4 to 6. Because of the constant linear treatment effect assumption (Assumption 2), we can write the potential outcome as:

$$Y_i(t) = Y_i(0) + \beta t = \mu_0 + \beta t + v_{i0},$$

where μ_0 is the mean of Y(0) in the population, and v_{i0} is its stochastic component. Now, the observed outcome can be written as:

$$y_{i}^{obs} = \mu_{0} + \beta t + v_{i0} = \mu_{0} + \beta t + E(v_{io} \mid \widetilde{X}) + [v_{i0} - E(v_{i0} \mid \widetilde{X})]$$

= $\mu_{0} + \beta t + E(v_{i0} \mid \widetilde{X}) + \xi_{io}.$

¹⁵ Since we do not have data on native children who attend grades 4 and 6, the *predicted* percentage of immigrants is calculated as:

Predicted number of immigrants in grade 5

Similarly, the predicted number of children in grade 5 is calculated as the sum of the predicted number of immigrants plus the actual number of natives in grade 5.

The error term in this equation, ξ_{i0} , is potentially correlated with the treatment variable. However, because Z is randomly assigned conditional on \widetilde{X} , it implies that $E(v_0 | Z, \widetilde{X}) = E(v_0 | \widetilde{X})$. Hence, Z is uncorrelated with ξ_{i0} and can be used as an instrument in estimating the above equation, for a given value of \widetilde{X} . In practice, we will stratify the sample by the *predicted* number of immigrants in grades 4 to 6, and estimate the equation by IV within each narrow cell, always controlling for a flexible function of the key conditioning variables.¹⁶

Balancing Tests for the Instrument

Table 6 presents tests for whether our instrument, the predicted percentage of immigrants in grade 5, is balanced with respect to the background characteristics of native students and schools. The table is analogous in structure to Table 4 and it shows that the predicted percentage of immigrants is not systematically correlated with any of the observable characteristics of natives or their schools. As in Table 4, the most striking result is the sharp contrast between the insignificant coefficients for the row averages (the second to last column) and the highly significant coefficients from the naïve OLS estimation (the last column). Overall, Table 6 shows that once we control for the predicted number of immigrants in grade 4 to 6 and the predicted number of children in grade 5, the predicted percentage immigrants in grade 5 is not correlated with any of the observable school or peer characteristics.

¹⁶ The IV estimation strategy is not directly comparable to the stratified OLS strategy for two reasons: first, the conditioning variables in the IV strategy are the *predicted* number of immigrants in grades 4 to 6 and the *predicted* number of children in grade 5, as opposed to the *actual* numbers of the OLS strategy. Second, we limit ourselves to schools where the *predicted* number of immigrants in grades 4 to 6 is between 1 and 40. This results in a slightly different sample of schools relative to the OLS strategy, where we included all schools where the *actual* number of immigrants was between 1 and 40. An alternative is to use exactly the same sample and stratification as in the OLS strategy, treat the key conditioning variables also as endogenous, and instrument for them using their predicted values. The results are very similar to the ones reported below, and are available from the authors upon request.

Instrumental Variable Results

The upper panel in Table 7 reports the IV estimates from regressions with no individual and school controls while the lower panel includes these controls. In general, relative to the stratified OLS results, the IV estimates point to a larger adverse effect of immigration on native students' outcomes. The individual cell coefficients are mostly insignificant, but they tend to point to an adverse effect of immigration on both matriculation rates (as in Table 5) and on dropout rates (in contrast to Table 5). The weighted average of the cell coefficients and the interacted model coefficients are similar to one another, and they both suggest that immigrants have a significant adverse effect on the dropout rate and the matriculation rate of their native peers. Furthermore, the size of the estimated effects is not negligible. The specification with all the control variables included (bottom panel) indicates that a 10 point increase in the percentage immigrants in 5th grade raises the dropout rate of native students by 1.4 percentage points and lowers the individual matriculation rate by 2.7-3.2 percentage points. For comparison purposes, one additional year of father's education raises matriculation rates by 1.5-1.9 percentage points; a one standard deviation change in the elementary school's socio-economic index raises matriculation rates by 3.4 percentage points; and a one standard deviation change in 1991 verbal test scores raises matriculation rates by 4.0 percentage points.

The finding that the IV results are larger and more significant than OLS suggests that the latter estimates may have been biased towards zero because of the endogenous holding-back of immigrants within a school and/or because of measurement error in the treatment variable. In particular, the IV results suggest that schools tend to place more immigrants in the grade that has stronger native students, thus causing OLS to be biased towards zero.¹⁷

¹⁷ In the previous version of this paper (Gould, Lavy, and Paserman (2004b)), we reported weak effects of immigrants on native outcomes using the OLS stratification, but stronger adverse effects on disadvantaged native students. The previous version did not use the IV strategy described above. Using the IV strategy, we find much stronger effect on the whole population as shown in this section, but we did not find differential effects on native students according to their socio-economic background.

6. Robustness Checks

We explore the robustness of our results to alternative specifications in Table 8. For ease of exposition, we present only the coefficient on the treatment variable in the fully interacted model, for both the stratified OLS and stratified IV specifications; the weighted average estimates are similar and are available upon request. The first two columns use a finer stratification based on 10 sub-samples, two more than the previous set of results. The results are essentially unchanged in magnitude and significance for both outcome variables, and for both the OLS and IV specifications. The second two columns stratify the samples along two dimensions: the total number of immigrants in grades 4 to 6 (4 intervals) and the total number of children in grade 5 (3 intervals), for a total of 12 different cells. The stratification along two dimensions aims to approximate more accurately the control function, but it necessitates a coarser stratification along the dimension of the number of immigrants in grades 4 to 6, meaning that within the cells there is greater variance in the propensity of schools to absorb immigrants. With this stratification, the coefficients tend to be slightly larger in magnitude and significance for the dropout rate, and significantly larger for the matriculation rate.

Finally, the last set of estimates uses the *number* of immigrants in grade 5 as the explanatory variable, rather than the percentage. The results are similar to those using the percentage immigrants, although the implied effects are a bit larger – six additional immigrants in grade 5^{18} lower the matriculation rate by up to 1.8 percentage points according to the OLS specification, and by 5.8 percentage points according to the IV specification. Overall, the robustness checks confirm the previous findings: once we account for the selectivity of immigrants both across schools and between grades within a given school, we find that native outcomes are negatively affected by the presence of immigrants.

As an additional test of our identification strategy, we run a series of "placebo" regressions, where the treatment variable is the percentage of immigrants in grades 1 and 2 in

¹⁸ Roughly a 10 percent increase in the percentage immigrants; see Table 4.

the 1993/1994 school year, rather than the percentage of immigrants in grade 5. If the results in Tables 5 and 7 were due to unobserved school characteristics, we would expect that the long-term outcomes of fifth graders would be equally affected by the percentage of immigrants in the earlier grades as by the percentage of immigrants in their own grade. This scenario is not supported by the results in Table 9: the percentage of immigrants in either first or second grade has no effect on the high school outcomes of fifth grade children, in either the stratified OLS or the IV specification. We interpret this as evidence that our results are not spuriously driven by unobserved school characteristics.

7. Other Results

Gender Specific Effects

In Table 10, we investigate whether immigrant boys and immigrant girls have a differential effect on the outcomes of native boys and girls. We estimate the fully interacted model as in Tables 5 and 7, but we now focus on two right hand side variables: the percentage of immigrant boys in 5^{th} grade (i.e., 100 times the number of immigrant boys divided by the total number of children in the grade), and the percentage of immigrant girls in 5^{th} grade. Interestingly, we find that the matriculation rate of native boys is affected exclusively by the percentage of immigrant boys, while it is not affected by the presence of immigrant girls. This result holds in both the stratified OLS and stratified IV specifications. On the other hand, the male dropout rate is not significantly affected by the percentage of immigrants of either gender. For girls, the results are somewhat mixed: the OLS specification indicates that native girls are negatively affected primarily by the percentage of immigrant boys in 5^{th} grade in both dropout rates (p-value of 0.103) and matriculation rates; the IV results suggest that the dropout rate of native girls is roughly equally affected by the presence of immigrant girls or boys (although neither coefficient is statistically significant), while the matriculation rate is affected primarily by the presence of immigrant girls or boys (although neither coefficient is statistically significant), while the matriculation rate is affected primarily by the presence of immigrant girls or boys (although neither coefficient is statistically significant), while the matriculation rate is affected primarily by the presence of immigrant girls or boys (although neither coefficient is statistically significant), while the matriculation rate is affected primarily by the presence of immigrant girls in 5^{th} grade. If the main mechanism

explaining the adverse effect of immigration on native outcomes is peer influence, and if this influence operates primarily within gender, we would have expected that natives would have been affected mostly by the presence of immigrants of their same gender. We find some evidence for this hypothesis, especially for boys, but the mixed results for girls suggest that other factors, such as the crowding out of limited school resources, could also be important.

Decomposing the Effect of Immigrants on the Matriculation Rate of Natives

The results show that the presence of a significant proportion of immigrants during primary school has an adverse effect on the dropout rate and matriculation rate of natives. However, these outcome variables are clearly related to one another – it is not possible to pass the matriculation exams if one has dropped out of high school. Therefore, it is natural to ask how much of the adverse effect of immigration on matriculation rates is due to higher dropout rates, and how much of the effect is due to lower matriculation rates conditional on not dropping-out. Table 11 decomposes the overall effect into these two components.

In particular, we decompose the effect of increasing the immigrant percentage in 5th grade from zero to 10 percent. Using the stratified IV estimate from Table 7, the overall effect of this increase is to reduce the probability of passing the matriculation exams by 3.16 percentage points. Over 80 percent of this reduction (2.88 percentage points, to be precise) is due to the lower probability of passing the matriculation exams conditional on not dropping-out, while the rest (0.99 percentage points) is due to the higher dropout rate. Therefore, most of the effect of immigration on matriculation rates seems to be due to the lower rate of passing the matriculation exam, rather than the lower rate of staying in school long enough to take the matriculation exam.

8. Conclusion

This paper represents one of the first attempts to study the consequences of natives and immigrants interacting in the same social and learning environment, while paying particular attention to issues of identification and causality. In addition, this paper breaks from the existing literature on peer effects within schools by looking at the long-term effects on natives in high school, rather than focusing on contemporaneous effects. After accounting for the endogenous placement of immigrants, both across schools and between grades within schools, we find that the percentage of immigrants in 5th grade has a substantial adverse effect on the dropout rate and high school achievements (matriculation rate) of native Israelis. These results stand in contrast with some of the recent estimates in the U.S. literature on desegregation, which found little or no impact of desegregation programs on the outcomes of white students (Angrist and Lang, 2004; Guryan 2004). One possible explanation for the larger estimated effects in our study is that the average number of immigrant children in each class in our sample is particularly large. Also, given that average class size in Israel is also relatively high, it is possible that the migration wave crowded out school resources that were already strained.

In many respects, these findings can be of general interest beyond the local Israeli context. The empirical episode studied here examines how immigrants from a relatively educated and skilled background from the former Soviet Union affect the human capital outcomes of natives. Despite the high levels of parental education, our analysis shows that the parents of these immigrants exhibit many troubling socio-economic patterns which could lead to a crowding out of school resources or generate other forms of negative peer effects on the outcomes of natives. The lessons learned from this analysis are becoming increasingly relevant to Western countries that are expected to absorb growing numbers of immigrants from Central and Eastern Europe (especially following the enlargement of the European

Union), and others who are debating whether to move toward a more skill-biased immigration policy.¹⁹

Our findings may also have implications for income inequality. The most important dividing line in schooling in the US is high school completion. In Israel and in many European countries, the crucial threshold is passing the matriculation exams, which opens the door to a college education. Angrist and Lavy (2004) estimated the returns to a year of schooling in Israel at about 8.3 percent, while holders of the Israeli matriculation certificate earn a further 24 percent. The effect of immigrants on the matriculation outcomes of natives may therefore have severe implications for the income of disadvantaged natives as well as for income inequality.

¹⁹ We should also keep in mind that the number of highly skilled immigrants in the West is already nonnegligible. In the United States in the year 2000, the fraction of recent immigrants with a college degree is slightly larger than that of natives (22.0 percent versus 20.9 percent), and in some states it is substantially larger. In Ohio, the fraction of immigrants with a college degree is 37.4 percent, versus 17.6 percent among natives; in Michigan, it is 33.2 percent among immigrants versus 17.2 among natives; in Pennsylvania, it is 33.1 percent among immigrants versus 19.0 among natives. (Source: 2000 Census of Population).

References

- Altonji, Joseph G., and Card, David. "The Effects of Immigration on the Labor Market Outcomes of Less-Skilled Natives." In *Immigration, Trade and Labor*, edited by John M. Abowd, and Richard B. Freeman. Chicago: University of Chicago Press, 1991.
- Angrist, Joshua D. and Lang, Kevin. "Does School Integration Generate Peer Effects? Evidence from Boston's Metco Program," *American Economic Review*, 94(5), December 2004, 1613-1634.
- Angrist, Joshua D. and Lavy, Victor. "Using Maimonides' Rule to Estimate the Effect of Class Size on Scholastic Achievement." *Quarterly Journal of Economics*, 114(2), May 1999, pages 533-75.
- Angrist, Joshua D. and Lavy, Victor "The Effect of High Stakes High School Achievement Awards Evidence from a Group Randomized Trial", Mimeo., Hebrew University, September, 2004.
- Betts, Julian R. "Educational Crowding Out: Do Immigrants Affect the Educational Attainment of American Minorities?", in Daniel S. Hamermesh and Frank D. Bean (Eds.), *Help or Hindrance? The Economic Implications of Immigration for African-Americans*, New York: Russell Sage Foundation, 1998.
- Betts, Julian R. and Fairlie, Robert W. "Does Immigration Induce 'Native Flight' from Public Schools into Private Schools." *Journal of Public Economics*, 87(5-6), May 2003, pp. 987-1012.
- Betts, Julian R. and Zau, Andrew. "Peer Groups and Academic Achievement: Panel Evidence from Administrative Data." Mimeo., UC San Diego, 2004.
- Boozer, Michael A. and Stephen E. Cacciola. "Inside the 'Black Box' of Project STAR: Estimation of Peer Effects Using Experimental Data." Economic Growth Center Discussion Paper 832, Yale University, June 2001.
- Borjas, George J. "Do Foreign Students Crowd Out Native Students from Graduate Programs?" NBER Working Paper 10349, 2004.
- Borjas, George J., Freeman, Richard B, and Katz, Lawrence. "Searching for the Effect of Immigration on the Labor Market", *American Economic Review*, 86(2), May 1996, pp. 246-251.
- Borjas, George J. "The Labor Demand Curve *Is* Downward Sloping: Reexamining the Impact of Immigration on the Labor Market." *Quarterly Journal of Economics*, November 2003.
- Burke, Mary A. and Sass, Tim R. "Classroom Peer Effects and Student Achievement." Mimeo., Florida State University, December 2004.
- Card, David. "The Impact of the Mariel Boatlift on the Miami Labor Market." *Industrial and Labor Relations Review*, 43(1), January 1990, pp. 245-257.
- Card, David. "Immigrant Inflows, Native Outflows, and the Local Labor Market Impacts of Higher Immigration." *Journal of Labor Economics*, 19(1), January 2001, pp. 22-64.
- Cohen-Goldner, Sarit and Paserman, M. Daniele. "Mass Migration to Israel and Natives' Transitions from Employment." *Industrial and Labor Relations Review*, forthcoming.
- Cohen-Goldner, Sarit and Paserman, M. Daniele. "The Dynamic Impact of Immigration on Natives' Labor Market Outcomes: Evidence from Israel." CEPR Discussion Paper 4640, September 2004.
- Edin, Per-Anders, Peter Fredriksson, and Olof Åslund, "Ethnic Enclaves and the Economic Success of Immigrants Evidence from a Natural Experiment," *Quarterly Journal of Economics*, CXVIII (2003), 329-357.
- Friedberg, Rachel M. "The Impact of Mass Migration on the Israeli Labor Market." *Quarterly Journal of Economics*, 116(4), November 2001, pp. 1373-1408.

- Gould, Eric D.; Lavy Victor and Paserman, M. Daniele. "Immigrating to Opportunity: Estimating the Effects of School Quality Using a Natural Experiment on Ethiopians in Israel." *Quarterly Journal of Economics*, May 2004a.
- Gould, Eric D.; Lavy Victor and Paserman, M. Daniele. "Does Immigration Affect the Long-Term Educational Outcomes of Natives? Quasi-Experimental Evidence." NBER Working Paper 10844, October 2004b.
- Guryan, Jonathan, "Desegregation and Black Dropout Rates," *American Economic Review*, 94(4), September 2004, 919-943.
- Hanushek, Eric; John F. Kain; Jacob M. Markman and Steven G. Rivkin. "Does Peer Ability Affect Student Achievement?" *Journal of Applied Econometrics*, 18(5), September/October 2003, pages 527-544.
- Hanushek, Eric, John F. Kain, and Steven G. Rivkin, "New Evidence About Brown V. Board of Education: The Complex Effects of School Racial Composition on Achievement," NBER Working Paper No. 8741, 2002.
- Heckman, James J. and Robb, Richard Jr. "Alternative Methods for Evaluating the Impact of Interventions: An Overview," in James J. Heckman and Burton Singer (Eds.), *Longitudinal Analysis of Labor Market Data*. Cambridge: Cambridge University Press, 1985.
- Hoxby, Caroline M. "Do Immigrants Crowd Disadvantaged American Natives Out of Higher Education?" in Daniel S. Hamermesh and Frank D. Bean (Eds.), *Help or Hindrance? The Economic Implications of Immigration for African Americans*. New York: Russell Sage Foundation, 1998.
- Hoxby, Caroline M. "Peer Effects in the Classroom: Learning from Gender and Race Variation." NBER Working Paper No. 7867, August 2000a.
- Hoxby, Caroline M. "The Effects of Class Size on Student Achievement: New Evidence from Population Variation." *Quarterly Journal of Economics*; 115(4), November 2000b, pages 1239-85.
- Hunt, Jennifer. "The Impact of the 1962 Repatriates from Algeria on the French Labor Market." *Industrial and Labor Relations Review*, 45(3), April 1992, pp. 556-572.
- Jacob, Brian A. "Public Housing, Housing Vouchers and Student Achievement: Evidence from Public Housing Demolitions in Chicago," *American Economic Review* 94(1), March 2004, 233-258.
- Jepsen, Christopher, and Steven Rivkin. "What is the Tradeoff between Smaller Classes and Teacher Quality?" NBER Working Paper No. 9205, September 2002.
- Katz, Lawrence F., Jeffrey R. Kling, and Jeffrey B. Liebman, "Moving to Opportunity in Boston: Early Results from a Randomized Mobility Experiment," *Quarterly Journal of Economics* CXVI (2001), 607-654.
- Krueger, Alan B. "Experimental Estimates of Education Production Functions", *Quarterly Journal of Economics*, CXIV (1999), 497-532.
- LaLonde, Robert and Topel, Robert. "Labor Market Adjustments to Increased Immigration." In *Immigration, Trade and Labor*, edited by John M. Abowd, and Richard B. Freeman. Chicago: University of Chicago Press, 1991.
- Lefgren, Lars. "Educational Peer Effects and the Chicago Public Schools." *Journal of Urban Economics*, 56(2), September 2004, pages 169-191.
- Oreopoulos, Philip, "The Long-Run Consequences of Living in a Poor Neighborhood," *Quarterly Journal of Economics*, CXVIII (2003), 1533-1575.
- Sacerdote, Bruce, "Peer Effects with Random Assignment: Results for Dartmouth Roommates," *Quarterly Journal of Economics*, CXVI (2001), 681-704.
- Vigdor, Jacob and Nechyba, Thomas. "Peer Effects in Elementary School: Learning from 'Apparent' Random Assignment." Mimeo., Duke University, October 2004.

- Weinberg, Bruce A., Patricia B. Reagan, and Jeffrey J. Yankow, "Do Neighborhoods Affect Hours Worked: Evidence from Longitudinal Data," *Journal of Labor Economics*, 2004, 22 (4), 891-924.
- Zimmerman, David J., "Peer Effects in Academic Outcomes: Evidence from a Natural Experiment," *Review of Economics and Statistics*, LXXXV (2003), 9-23.

A: Ministry of Ed	ucation data:				
	All	Ir	nmigrants from:		
	Immigrants	Former Soviet Union Ethiopia		Other	Native Israelis
Father's years of schooling	11.45 (5.49)	13.04 (3.62)	1.80 (3.30)	14.21 (4.46)	12.03 (3.56)
Mother's years of schooling	11.58 (5.31)	13.25 (3.26)	1.39 (2.90)	13.72 (3.70)	12.06 (3.41)
Number of siblings	1.54 (1.73)	1.11 (1.16)	4.28 (2.61)	1.98 (1.35)	2.41 (1.57)
Number of children	7,159	5,477	945	737	54,700

Table 1: Background	Characteristics o	of Immigrants	and Natives
---------------------	--------------------------	---------------	-------------

B: Census 1995 da	ata				
	All	I	mmigrants from		
	Immigrants	Former Soviet Union	Ethiopia	Other	Native Israelis
Father's years of	12.69	13.51	2.57	13.50	11.67
schooling	(4.50)	(3.20)	(4.61)	(4.32)	(3.83)
Mother's years	12.53	13.45	1.92	13.40	11.59
of schooling	(4.52)	(3.09)	(4.20)	(3.95)	(3.98)
Female headed	0.176	0.179	0.248	0.093	0.085
household	(0.381)	(0.383)	(0.432)	(0.290)	(0.278)
Home	0.533	0.561	0.482	0.338	0.693
ownership	(0.499)	(0.496)	(0.500)	(0.473)	(0.461)
Household head	0.071	0.067	0.096	0.086	0.037
unemployed	(0.256)	(0.252)	(0.295)	(0.281)	(0.189)
Household	3895.85	4258.76	751.01	3497.91	5899.36
monthly income	(3911.46)	(3738.91)	(1184.43)	(5371.44)	(6963.88)
Number of households	8,910	7,321	737	852	68,725

Notes: The Ministry of Education sample includes the universe of children enrolled in 5th grade during the 1993-1994 school year. The Census data includes all households with at least one child between the ages of 8 and 12 in 1995.

	Number of immigrants in grades 4 through 6									
	1-10	11-20	21-30	31-40	All (1-40)					
Number of schools	271	244	164	124	803					
Total number of children in grade 5	12,593	13,175	9,573	7,005	42,346					
Average number of children in grade 5	50.49	62.02	69.96	71.44	61.21					
Average percentage immigrants in grade 5	5.39	10.22	13.97	19.48	10.79					
Peer characteristics (non-immigrants)										
Father's years of	12.62	12.24	11.97	11.91	12.24					
schooling	(3.66)	(3.53)	(3.32)	(3.34)	(3.50)					
Mother's years of	12.65	12.23	12.04	11.83	12.25					
schooling	(3.47)	(3.37)	(3.13)	(3.29)	(3.35)					
c	2.43	2.45	2.37	2.31	2.40					
Number of siblings	(1.69)	(1.54)	(1.52)	(1.42)	(1.57)					
	0.240	0.290	0.317	0.312	0.285					
Fraction Asia-Africa	(0.427)	(0.454)	(0.465)	(0.463)	(0.451)					
	30.08	31.09	32.49	33.04	31.43					
Class Size	(6.04)	(5.74)	(5.16)	(5.36)	(5.74)					
School characteristics										
Standardized Math	0.347	0.164	0.115	0.102	0.197					
Score in 1991	(0.772)	(0.837)	(0.802)	(0.814)	(0.813)					
Standardized Verbal	0.410	0.144	0.023	0.011	0.174					
Score in 1991	(0.769)	(0.799)	(0.739)	(0.808)	(0.795)					
Socioeconomic Index	0.397	0.194	0.155	0.129	0.235					
in 1991	(0.749)	(0.817)	(0.811)	(0.814)	(0.803)					
Outcome Variables										
Fraction dropping out	0.047	0.055	0.059	0.058	0.054					
before completing 12 th grade	(0.211)	(0.229)	(0.236)	(0.233)	(0.226)					
Fraction eligible for										
high school	0.642	0.611	0.605	0.584	0.614					
matriculation	(0.480)	(0.488)	(0.489)	(0.493)	(0.487)					

Table 2: Summary Statistics

Notes: Authors' calculations based on Ministry of Education data

	(1)	(2)	(3)
Dependent variable:			
Dropped out before completing 12 th grade	0.0013 [6.07]	0.0004 [1.71]	0.0003 [0.95]
Passed HS matriculation exam	-0.0057 [-8.75]	-0.0013 [-2.68]	-0.0014 [-2.03]
Controls for individual and school characteristics	No	Yes	Yes
Controls for number of immigrants in grades 4 to 6 and number of children in grade 5	No	No	Yes

Table 3: Naïve OLS estimates of the effect of immigration on natives' long-term educational outcomes

Note: Entries in the table represent the coefficients from separate regressions of the relevant dependent variable on the percentage immigrants in 5th grade. Individual controls: mother's years of schooling, father's years of schooling, dummy indicators for whether parents' schooling is missing, number of siblings, a dummy for whether the number of siblings is missing, age in 1994, a gender dummy. School controls: a dummy for whether the elementary school is religious, class size in 1994, math scores in 1991, verbal scores in 1991, and school socioeconomic index in 1991. Robust t-statistics (adjusted for clustering at the school level) in brackets.

			Nur	nber of immig	rants in grades	s 4-6				Naïve OLS
	1-5	6-10	11-15	16-20	21-25	26-30	31-35	36-40	Row average	
Dependent Variable:										
Mother's years of	0.0625	0.0054	0.0118	0.0115	0.0293	-0.0244	-0.0299	-0.0357	-0.0001	-0.0526
schooling	[1.34]	[0.16]	[0.42]	[0.36]	[1.13]	[-1.05]	[-0.92]	[-1.06]	[-0.01]	[-7.79]
Father's years of	0.0640	0.0085	0.0193	0.0040	0.0276	-0.0364	-0.0222	-0.0587	-0.0026	-0.0540
schooling	[1.27]	[0.24]	[0.70]	[0.13]	[1.08]	[-1.50]	[-0.60]	[-1.66]	[-0.24]	[-7.81]
Number of	-0.0057	0.0374	-0.0314	0.0197	-0.0130	0.0224	0.0133	0.0050	0.0066	0.0125
Siblings	[-0.18]	[2.17]	[-2.08]	[1.22]	[-0.66]	[1.47]	[0.59]	[0.28]	[1.04]	[3.01]
Fraction	-0.0014	-0.0001	-0.0028	0.0034	-0.0001	0.0021	0.0001	-0.0075	-0.0006	0.0053
Asia-Africa	[-0.33]	[-0.04]	[-0.92]	[1.05]	[-0.02]	[0.68]	[0.03]	[-2.15]	[-0.54]	[7.65]
Standardized math	0.0077	-0.0147	-0.0217	-0.0151	0.0117	-0.0024	-0.0180	0.0069	-0.0064	-0.0317
score in 1991	[0.37]	[-0.74]	[-1.14]	[-1.08]	[0.68]	[-0.13]	[-0.96]	[0.36]	[-1.00]	[-7.93]
Standardized verbal	0.0133	-0.0095	0.0048	-0.0196	0.0037	-0.0034	-0.0168	0.0127	-0.0042	-0.0348
score in 1991	[0.60]	[-0.51]	[0.24]	[-1.45]	[0.22]	[-0.21]	[-0.95]	[0.70]	[-0.68]	[-9.03]
Socioeconomic	0.0012	-0.0200	0.0157	-0.0219	-0.0187	-0.0181	0.0003	0.0162	-0.0068	-0.0379
index in 1991	[0.05]	[-1.04]	[0.88]	[-1.54]	[-0.99]	[-0.93]	[0.02]	[0.84]	[-1.06]	[-10.14]
Class size in	-0.1709	-0.2134	0.2698	0.0244	0.0838	0.0778	0.0186	0.0689	0.0222	-0.0657
1994	[-1.33]	[-2.37]	[2.70]	[0.25]	[0.83]	[0.75]	[0.23]	[0.73]	[0.64]	[-2.48]
Number of schools	123	148	153	91	99	65	72	52		803
Number of children	4,872	7,721	8,053	5,122	5,610	3,963	3,959	3,046	42	2,346
Average number of children in 5 th grade	42.20	57.38	59.46	66.34	67.85	73.17	69.50	74.13	6	1.20
Average number of immigrants in 5 th grade	1.04	3.06	4.39	6.56	7.59	9.65	11.14	12.40	4	5.82
Average Percentage Immigrants in 5 th grade	3.48	6.97	9.05	12.19	13.35	14.93	19.04	20.09	1	0.79
Standard deviation of % Immigrants in 5 th grade	4.73	5.84	5.98	9.70	7.89	7.75	9.88	11.18	9	9.03

Table 4: Balancing Tests for the Actual Percentage of Immigrants

Notes: Entries in the table represent the coefficients from separate regressions of the relevant dependent variable on the percentage immigrants in 5th grade. All regressions control for number of immigrants in grades 4 to 6, number of children in grade 5, their squares, and the interaction between the two. Robust t-statistics, adjusted for clustering at the school level, in brackets.

				variable: Perce	0 0	0				
	Estir	nation method:		ased on number	Ŭ	<u> </u>	5, and OLS with	nin cells	Weighted	
	Number of immigrants in grades 4 to 6:									Interacted
	1-5	6-10	11-15	16-20	21-25	26-30	31-35	36-40	Average	Model
A: No Individual Controls										
Dropped out before completing 12 th grade	-0.0022 [-1.23]	0.0012 [1.33]	-0.0003 [-0.29]	0.0013 [1.84]	0.0003 [0.31]	0.0009 [0.98]	0.0016 [1.66]	-0.0017 [-1.25]	0.0006 [1.84]	0.0003 [0.88]
Passed HS matriculation exam	0.0031 [0.66]	-0.0016 [-0.49]	-0.0024 [-0.80]	-0.0018 [-0.66]	0.0006 [0.23]	-0.0039 [-1.31]	-0.0063 [-1.97]	0.0020 [0.63]	-0.0015 [-1.37]	-0.0016 [-1.46]
B: With Individual Controls										
Dropped out before completing 12 th grade	-0.0023 [-1.45]	0.0011 [1.37]	0.0003 [0.31]	0.0010 [1.40]	0.0003 [0.37]	0.0003 [0.43]	0.0008 [0.91]	-0.0012 [-0.96]	0.0004 [1.35]	0.0003 [0.77]
Passed HS matriculation exam	0.0002 [0.04]	-0.0016 [-0.64]	-0.0040 [-1.60]	-0.0007 [-0.37]	-0.0012 [-0.70]	-0.0019 [-0.77]	-0.0033 [-1.89]	0.0015 [0.76]	-0.0014 [-1.88]	-0.0015 [-1.88]
Number of schools	123	148	153	91	99	65	72	52	8	03
Number of children	4,872	7,721	8,053	5,122	5,610	3,963	3,959	3,046	42,	,346

Table 5: Stratified OLS Regressions for the Outcomes for Native Israelis – Accounting for Endogenous Sorting Across Schools	5
Treatment variable: Percentage immigrants in grade 5	

Notes: Entries in the first eight columns represent the estimated effect of percentage immigrants in 5th grade on the relevant dependent variable, within narrow cells defined by the actual number of immigrants in grades 4 to 6. The last two columns represent summary measures of the estimated effect (see text for details). All regressions control for the number of children in grade 5, the number of immigrants in grades 4 to 6, their squares, and the interaction between the two. Individual controls are mother's years of schooling, father's years of schooling, dummy indicators for whether parents' schooling is missing, number of siblings, a dummy for whether the number of siblings is missing, age in 1994, a gender dummy, a dummy for whether the elementary school is religious, class size in 1994, math scores in 1991, verbal scores in 1991, and school socioeconomic index in 1991. Robust t-statistics (adjusted for clustering at the school level) in brackets.

			Predicted	number of im	migrants in gra	ades 4 to 6				
-	1-5	6-10	11-15	16-20	21-25	26-30	31-35	36-40	Row average	Naïve OLS
Mother's years of schooling	-0.0245	0.0101	0.0270	0.0236	-0.0066	-0.0052	0.0107	-0.0835	0.0008	-0.0559
	[-0.54]	[0.24]	[0.90]	[0.65]	[-0.23]	[-0.20]	[0.40]	[-1.71]	[0.07]	[-8.45]
Father's years of schooling	-0.0066	0.0209	0.0155	0.0347	-0.0097	-0.0037	0.0346	-0.1025	0.0052	-0.0553
	[-0.13]	[0.45]	[0.51]	[0.96]	[-0.35]	[-0.14]	[1.29]	[-2.10]	[0.44]	[-8.17]
Number of	0.0746	0.0291	-0.0156	0.0149	-0.0023	-0.0034	0.0287	-0.0070	0.0112	0.0109
Siblings	[2.64]	[1.47]	[-0.71]	[0.95]	[-0.10]	[-0.22]	[1.55]	[-0.36]	[1.65]	[2.82]
Fraction	0.0112	0.0012	-0.0013	-0.0016	-0.0044	0.0007	0.0001	0.0024	0.0005	0.0057
Asia-Africa	[2.41]	[0.33]	[-0.35]	[-0.44]	[-1.14]	[0.22]	[0.04]	[0.52]	[0.39]	[8.04]
Standardized math score in 1991	0.0093	-0.0059	-0.0287	-0.0084	0.0007	-0.0005	0.0000	-0.0563	-0.0115	-0.0318
	[0.38]	[-0.26]	[-1.69]	[-0.55]	[0.04]	[-0.03]	[0.00]	[-2.65]	[-1.72]	[-7.81]
Standardized verbal score in 1991	-0.0059	-0.0115	-0.0113	-0.0086	0.0040	0.0034	-0.0047	-0.0518	-0.0098	-0.0354
	[-0.25]	[-0.49]	[-0.64]	[-0.64]	[0.19]	[0.19]	[-0.30]	[-2.52]	[-1.53]	[-9.28]
Socioeconomic index in 1991	-0.0419	-0.0181	-0.0125	-0.0061	-0.0125	-0.0117	0.0043	-0.0031	-0.0102	-0.0379
	[-1.61]	[-0.84]	[-0.73]	[-0.33]	[-0.56]	[-0.59]	[0.27]	[-0.13]	[-1.44]	[-10.28]
Class size in	0.0052	-0.1488	0.0140	0.1972	-0.1415	-0.0438	-0.1894	-0.1002	-0.0395	-0.0868
1994	[0.04]	[-1.34]	[0.12]	[2.21]	[-1.45]	[-0.52]	[-1.65]	[-0.86]	[-1.07]	[-3.34]
Number of schools	113	158	136	93	89	86	64	61	80	0
Number of children	4,461	8,209	7,170	5,058	5,186	4,701	3,859	3,687	42,3	31
Average number of children in 5 th grade	42.17	57.12	59.58	63.77	68.54	67.15	74.84	75.39	61	36
Average number of immigrants in 5 th grade	1.09	3.01	4.50	6.06	7.13	9.08	11.17	11.97	5.8	0
Average Percentage Immigrants in 5 th grade	3.64	6.96	9.13	11.69	12.67	15.41	18.14	17.79	10.0	57
Standard deviation of % Immigrants in 5 th grade	5.03	6.10	6.59	10.00	8.32	7.72	11.15	8.35	8.9	5

Table 6: Balancing Tests for the Predicted Percentage of Immigrants

Notes: Entries in the table represent the coefficients from separate regressions of the relevant dependent variable on the predicted percentage immigrants in 5th grade. The predicted percentage is the percentage of immigrants that should have been in 5th grade based on their dates of birth. All regressions control for predicted number of immigrants in grades 4 to 6, predicted number of children in grade 5, their squares, and their interaction. Robust t-statistics, adjusted for clustering at the school level, in brackets.

	Eat	impetion mothed			entage immigrat		6 and W with	in colle		
	ESt	imation method			entage immigration	0	o, and i v with	in cens		
Predicted number of immigrants in grades 4 to 6:										Interacted
	1-5	6-10	11-15	16-20	21-25	26-30	31-35	36-40	Average	Model
A: No individual or school controls										
Dropped out before completing 12 th grade	0.0035 [1.37]	0.0012 [0.63]	0.0017 [1.15]	0.0021 [1.92]	0.0001 [0.03]	0.0023 [1.08]	0.0013 [0.68]	0.0013 [0.72]	0.0017 [2.84]	0.0016 [2.59]
Passed HS matriculation exam	0.0065 [0.78]	-0.0049 [-0.77]	-0.0028 [-0.55]	0.0010 [0.29]	-0.0098 [-1.27]	0.0005 [0.08]	-0.0001 [-0.02]	-0.0156 [-2.69]	-0.0024 [-1.27]	-0.0032 [-1.62]
B: With individual and school controls										
Dropped out before completing 12 th grade	0.0028 [1.18]	0.0012 [0.67]	0.0021 [1.48]	0.0022 [1.72]	-0.0006 [-0.27]	0.0027 [1.62]	0.0010 [0.57]	-0.0014 [-0.66]	0.0015 [2.51]	0.0014 [2.23]
Passed HS matriculation exam	0.0078 [1.10]	-0.0056 [-1.26]	-0.0041 [-1.17]	-0.0014 [-0.51]	-0.0106 [-1.89]	-0.0001 [-0.03]	-0.0007 [-0.19]	-0.0074 [-1.66]	-0.0028 [-2.00]	-0.0032 [-2.12]
Number of schools Number of children	113 4461	158 8209	136 7170	93 5058	89 5186	86 4701	64 3859	61 3687		00 331

 Table 7: Stratified IV Regressions for Outcomes for Native Israelis – Accounting for Endogeneity in Grade Placement

Notes: Entries in the first eight columns represent the estimated effect of percentage immigrants in 5^{th} grade on the relevant dependent variable, within narrow cells defined by the predicted number of immigrants in grades 4 to 6. The last two columns represent summary measures of the estimated effect (see text for details). All regressions control for the predicted number of children in grade 5, the predicted number of immigrants in grades 4 to 6, their squares, and the interaction between the two. Individual controls are mother's years of schooling, father's years of schooling, dummy indicators for whether parents' schooling is missing, number of siblings, a dummy for whether the number of siblings is missing, age in 1994, a gender dummy, a dummy for whether the elementary school is religious, class size in 1994, math scores in 1991, verbal scores in 1991, and school socioeconomic index in 1991. Robust t-statistics (adjusted for clustering at the school level) in brackets.

	(1) Finer St	ratification	(2) Two-dimensio	nal stratification	(3) Number o	f immigrants	
	Treatment variable: Perc	entage immigrants in	Treatment variable: Perc	entage immigrants in	Treatment variable: Number of immigrants in		
	grade 5		grade 5		grade 5.		
	Stratification: Actual/Pro		Stratification: Actual/Pre		Stratification: Actual/pre		
		grades 4 to 6 (10 cells)	immigrants in g			grades 4 to 6 (8 cells)	
	Instrument: Predicted pe	5 5	1	number of children in	Instrument: Predicted nu	U U	
	grade 5 (IV onl	y)	grade 5 (12 cells		grade 5 (IV only)		
			Instrument: Predicted pe				
			grade 5 (IV onl	y)			
	Stratified OLS	Stratified IV	Stratified OLS	Stratified IV	Stratified OLS	Stratified IV	
Dependent variable:							
Dropped out before	0.0002	0.0014	0.0002	0.0018	0.0009	0.0022	
completing 12 th grade	[0.57]	[2.50]	[0.54]	[2.60]	[1.40]	[1.65]	
Passed HS	-0.0014	-0.0028	-0.0023	-0.0052	-0.0032	-0.0098	
matriculation exam	[-1.77]	[-1.97]	[-2.67]	[-2.93]	[-2.37]	[-2.94]	

Table 8: Robustness Tests

Note: Entries in the table represent the estimated effect of the treatment variable on the relevant dependent variable from separate models. The effect is calculated based on the fully interacted model. All regressions control for predicted number of children in grade5, the predicted number of immigrants in grades 4 to 6, their squares, and the interaction between the two. Individual controls are mother's years of schooling, father's years of schooling, dummy indicators for whether parents' schooling is missing, number of siblings, a dummy for whether the number of siblings is missing, age in 1994, a gender dummy, a dummy for whether the elementary school is religious, class size in 1994, math scores in 1991, verbal scores in 1991, and school socioeconomic index in 1991. Robust t-statistics (adjusted for clustering at the school level) in brackets.

Table 9: Placebo Regressions

	Estimation method: S	DLS – Fully intera Stratification based on r es 4 to 6, and OLS with	umber of immigrants	Stratified IV – Fully interacted Model Estimation method: Stratification based on number of immigrants in grades 4 to 6, and IV within cells (instrument: predicted percentage immigrants in grade 5)					
		Treatment variable	:		Treatment variable:				
	True effect:	"Placebo":	Placebo:	True effect:	"Placebo":	Placebo:			
	% immigrants in	% immigrants in	% immigrants in	% immigrants in	% immigrants in	% immigrants in			
	5 th grade	1 st grade	2 nd grade	5 th grade	1 st grade	2 nd grade			
Dropped out before completing 12 th grade	0.0003	-0.0000	-0.0000	0.0014	-0.0000	-0.0001			
	[0.77]	[-0.27]	[-0.74]	[2.23]	[-0.35]	[-1.50]			
Passed HS matriculation	-0.0015	0.0000	0.0001	-0.0032	0.0001	0.0002			
exam	[-1.88]	[0.27]	[0.92]	[-2.12]	[0.65]	[1.20]			

Notes: Entries in the table represent the coefficients on the treatment variable in the fully interacted model (see text for details). All regressions control for the predicted number of children in grade 5, the predicted number of immigrants in grades 4 to 6, their squares, and the interaction between the two. Individual controls are mother's years of schooling, father's years of schooling, dummy indicators for whether parents' schooling is missing, number of siblings, a dummy for whether the number of siblings is missing, age in 1994, a gender dummy, a dummy for whether the elementary school is religious, class size in 1994, math scores in 1991, verbal scores in 1991, and school socioeconomic index in 1991. Robust t-statistics (adjusted for clustering at the school level) in brackets.

		Males Fem		ales	
		Stratified OLS	Stratified IV	Stratified OLS	Stratified IV
Dependent variable: Dropped out before completing 12 th grade	Treatment variable:				
	Percentage	-0.0003	0.0017	0.0009	0.0015
	Immigrant Males	[-0.37]	[1.23]	[1.63]	[1.50]
	Percentage	-0.0002	0.0002	0.0006	0.0011
	Immigrant Females	[-0.25]	[0.16]	[0.98]	[1.26]
Dependent variable: Passed HS matriculation exam	Percentage Immigrant Males	-0.0030 [-2.01]	-0.0057 [-2.05]	-0.0032 [-2.15]	-0.0026 [-0.93]
	Percentage	0.0011	-0.0009	-0.0006	-0.0039
	Immigrant Females	[0.82]	[-0.37]	[-0.40]	[-1.76]
Number of schools		779	777	776	775
Number of schools		20,766	20,790	20,867	20,842

Table 10: Gender Specific Effects

Notes: Entries in the table represent the coefficients on the treatment variable in the fully interacted model (see text for details). All regressions control for the predicted number of children in grade 5, the predicted number of immigrants in grades 4 to 6, their squares, and the interaction between the two. Individual controls are mother's years of schooling, father's years of schooling, dummy indicators for whether parents' schooling is missing, number of siblings, a dummy for whether the number of siblings is missing, age in 1994, a gender dummy, a dummy for whether the elementary school is religious, class size in 1994, math scores in 1991, verbal scores in 1991, and school socioeconomic index in 1991. Robust t-statistics (adjusted for clustering at the school level) in brackets.

Table 11: Decomposing the IV Effect of Immigrants on the Matriculation Rate of Natives						
(1)	(2)	Difference between Columns (1) and (2)				
Prob (HS matriculation pct. Immigrants = 0%)	Prob (HS matriculation pct. Immigrants = 10%)					
0.5766	0.5451	0.0316				
Prob (HS matriculation not dropping out, pct. Immigrants = 0%)	Prob (HS matriculation not dropping out, pct. Immigrants = 10%)		% of total diff. explained by diff.			
0.6185	0.5897	0.0288	in conditional matriculation rat 81.6%			
Prob (not dropping out pct. Immigrants = 0%)	Prob (not dropping out pct. Immigrants = 10%)		% of total diff.			
0.9276	0.9177	0.0099	explained by diff in probability of no dropping out: 18.4%			

Notes: The conditional probabilities are calculated for a male native student who attended a non-religious elementary school with 16 immigrants in grades 4 to 6 and 73 5th grade children altogether, whose fifth grade class had 32 children, whose parents were born in Israel and both completed 12 years of education, who has two siblings, who was 11 years old when he attended 5th grade, and whose elementary school had the national average in terms of 1991 socioeconomic status, math test scores and reading test scores.